Author's response to reviews

Title: The Transition of Reported Pain in Different Body Regions - A One-Year Follow-Up Study

Authors:

Christina Gummesson (christina.gummesson@med.lu.se)
Sven-Olof Isacsson (sven-olof.isacsson@med.lu.se)
Agneta H Isacsson (sven-olof.isacsson@med.lu.se)
H.Ingemar Andersson (ingemar.andersson@hv.hkr.se)
John Ektor-Andersen (john.ektor-andersen@skane.se)
Per-Olof Ostergren (per-olof.ostergren@med.lu.se)
Bertil Hanson (sven-olof.isacsson@med.lu.se)
the Malmo Shoulder-Neck Study group (per-olof.ostergren@med.lu.se)

Version: 7 Date: 10 February 2006

Author's response to reviews:

To the Editor,

We have now revised the format as indicated below.

Author list -
The problem with the list in the submission system is now resolved so that the author list is as in the manuscript. However, the e-mail list is not exactly the same, but done as advised by the editorial production team (one author should not be assigned any e-mail address in the manuscript).

Section headings -
ABSTRACT is changed to Abstract

References -
All references are before the full stop.

Figure legends -
The figure legends are placed after the references and before the tables.

References -
We have now changed in the biomedcentral output style for RefMan so that the reference list contains all authors’ names.

Tables -
The tables are now submitted in word format so that the format is visible. Visible black lines are added to the borders of the tables.

We hope that the format changes made are satisfactory. We would be happy to address any further comments.

Yours sincerely,
Christina Gummesson